Author's response to reviews

Title: Knowledge of the abortion legislation among South African women: a cross-sectional study

Authors:

Chelsea Morroni (chelsea@cormack.uct.ac.za)
Landon Myer (lmyer@cormack.uct.ac.za)
Kemilembe Tibazarwa (tbzber001@mail.uct.ac.za)

Version: 5 Date: 29 July 2006

Author's response to reviews:

29 July 2006

Please find our revised manuscript enclosed (Knowledge of the abortion legislation among South African women: a cross-sectional study). We have made all of the formatting changes requested.

Formatting changes requested

Underlining - Please do not underline any section headings. DONE

Authors' contributions - Please indicate that all authors read and approved the final manuscript. DONE

References - The reference list should contain all authors' names, if there are 30 or below. The term 'et al.' should not be used. DONE

References - Please write all page numbers in full: e.g. 223-227 NOT 223-7. DONE

Yours sincerely

Chelsea Morroni
Senior Researcher, Women's Health Research Unit